Author's response to reviews

Title: The Diagnostic Yield of the first episode of a Periodic Health Evaluation

Authors:

Cindy A Kermott (kermott.cindy@mayo.edu)
Carol S Kuhle (kuhle.carol@mayo.edu)
Stephanie S Faubion (faubion.stephanie@mayo.edu)
Ruth E Johnson (johnsonmd.ruth@mayo.edu)
Donald D Hensrud (hensrud.donald@mayo.edu)
Mohammad H Murad (murad.mohammad@mayo.edu)

Version: 4 Date: 16 April 2012

Author's response to reviews: see over
Editors:
Final recommended edit change made and revised manuscript has now been uploaded. Thank you very much.
Cindy Kermott and co-authors.

MS: 2070445436640948
Research article
The Diagnostic Yield of the first episode of a Periodic Health Evaluation
Cindy A Kermott, Carol S Kuhle, Stephanie S Faubion, Ruth E Johnson, Donald D Hensrud and Mohammad H Murad
BMC Health Services Research (Section: Quality, performance, safety and outcomes)

Dear Dr. Kermott:

We are pleased to inform you that the editors have now decided to accept your manuscript. However, before we can proceed we will need you to make some changes to your manuscript. We strongly urge you to make these changes promptly, as we cannot proceed to the next process until we have received a version containing the changes.

Associate Editor’s comments:

The only remaining error I found in table 1 is that there is not a lower level for the youngest age range (currently says -39)

Done.

Once you have made these changes, please upload the revised version by following these instructions:

Done.